Author’s response to reviews

Title: Cryptic FMR1 mosaic deletion in a phenotypically normal mother of a boy with Fragile X Syndrome: case report

Authors:

Shiyu Luo (luoshiyu@sklmg.edu.cn)
Wen Huang (huangwen@sklmg.edu.cn)
Qiuping Xia (xiqiuping@sklmg.edu.cn)
Yan Xia (xiayan@sklmg.edu.cn)
Qian Du (duqian@sklmg.edu.cn)
Lingqian Wu (wulingqian@sklmg.edu.cn)
Ranhui Duan (duanranhui@sklmg.edu.cn)

Version: 3
Date: 2 November 2014

Author’s response to reviews: see over
Dear Editors,

Many thanks for handling our manuscript. We followed the suggestion and added a permission-to-publish statement in the 'Consent' section.

Minor essential revisions

-Please revise the 'Consent' section to include to add permission to publish statement. We recommend the following wording is used for the consent section: "Written informed consent was obtained from the mother of the patient for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal."

Answer:

1) The consent for the publication was obtained from the mother of the patient as stated in the Consent section on page 11.

Original: “Informed written consent for the collection of tissue samples and subsequent analyses was obtained from the mother. A copy of the written consent is available for review by the Editor of this Journal.”

Revised: “Written informed consent was obtained from the mother of the patient for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal.”

Thank you very much for your attention and consideration. If any parts of the manuscript need further improvement, please inform us for revision. I'm looking forward to hearing from you soon.

With kind regards,
Sincerely,
Ranhui Duan, Ph.D.
Professor
The State Key Lab of Medical Genetics,
Central South University,
Changsha, 410078, Hunan,
P. R. China.
Tel: +8673184805349;
Fax: +8673184805349;
E-mail: duanranhui@sklmg.edu.cn